Author's response to reviews

Title: Early and late mortality in elderly patients after hip fracture: a cohort study using administrative health databases in the Lazio region, Italy

Authors:

Esmeralda Castronuovo (Castronuovo@asplazio.it)
Patrizio Pezzotti (Pezzotti@asplazio.it)
Antonella Franzo (antonella.franzo@sanita.fvg.it)
Domenico Di Lallo (Dilallo@asplazio.it)
Gabriella Guasticchi (Guasticchi@asplazio.it)

Version: 2 Date: 25 January 2011

Author's response to reviews: see over
24 January 2011

To the editors
BMC Geriatrics

RE: Manuscript entitled “Early and late mortality in elderly patients after hip fracture: a cohort study using administrative health databases in the Lazio region, Italy”

Dear Editors,

I have submitted the above-specified manuscript. The manuscript provides important information on the risk of dying after hip fracture in a population-based cohort of subjects residents in Lazio Region (central Italy) aged ≥65 years. In particular we furthermore, it provides the evaluating about roles of the determinants for either short term or long term fatal outcome. The manuscript has been written in accordance with the submission instructions of the journal.

All authors contributed to the planning of the study, the analysis, the editing of the manuscript; the final version of the manuscript was approved by all authors. I declare that there are no conflicts of interest of any author with the conclusions reported in this manuscript. The study design is in agreement with the international ethical principles of the scientific research, as well as the reported results.

I hope that the rationale, the study design, the statistical analysis, the conclusion and the style are suitable for the publication in BMC Geriatrics of this manuscript. Thank you in advance for your consideration.

Sincerely,

Esmeralda Castronuovo
Lazio Sanità – Agenzia di Sanità Pubblica della Regione Lazio
Via di Santa Costanza, 53
00198 Roma, Italy